Author's response to reviews

Title: Effect of short-acting beta blocker on the cardiac recovery after cardiopulmonary bypass

Authors:

jie sun (dgsunjie@hotmail.com)
zhengnian ding (jitazi0100@sina.com)
yanning qian (yanning_qian@yahoo.com.cn)

Version: 3 Date: 12 August 2011

Author's response to reviews:

1. We have made the title identical in the manuscript and the submission system.
2. We changed the authors' name format and their contributions according to your requirements.
3. We also changed the figure format.
4. We removed all the highlighting and changed "introduction" to "background".